Author's response to reviews

Title: Pudendal nerve decompression in perineology: a case series.

Authors:

Jacques Beco (jacques.beco@skynet.be)
Daniela Climov (dclimov@parnasse-deuxalice.edu)
Michele Bex (michele.bex@skynet.be)

Version: 9 Date: 19 October 2004

Author's response to reviews:

Sorry for this last correction but it is very important:

This sentence has been introduced page 14:

Therefore, if the "clamp" must be open efforts should be done to open it without cutting a ligament